Author's response to reviews

Title: Surgery is more cost-effective than splinting for carpal tunnel syndrome in the Netherlands: Results of an economic evaluation alongside a randomized controlled trial

Authors:

Ingeborg B.C. Korthals-de Bos (i.korthals@voorankerhuizen.nl)
Annette A.M. Gerritsen (a.gerritsen@vumc.nl)
Maurits W. van Tulder (mw.vantulder@vumc.nl)
Maureen P.M.H. Rutten-van Molken (mp.rutten@erasusmc.nl)
Herman J. Ader (h.ad@vumc.nl)
Henrica C.W. de Vet (hcw.devet@vumc.nl)
Lex M. Bouter (lm.bouter@vumc.nl)

Version: 4 Date: 15 November 2006

Author's response to reviews:

We have included the formatting changes as requested.
1. We have removed the highlighting from the text.
2. We have removed the full stop from the end of the title and included a colon in the middle of the title in stead of a full-stop.
3. We have removed the qualifications and job titles from the manuscript.
4. We have changed capitals into sentence case for all headings and subheadings.
5. We have included the trial registration number at the end of the abstract.
6. We have removed the keywords section.
7. We have changed the references from superscripts to numbers in square brackets throughout the text.
8. We have placed the references before the full stop.
9. We have removed the website from the corresponding author details.
10. WE have removed the figure and table placement from the text.
11. We noticed that some references were not cited in the final text and have removed these from the reference list. We have changed the references numbers accordingly in the text.
12. WE have modified the competing interests section.
13. WE have modified the section with authors' contribution.
14. WE have deleted the figure titles and legend from the files and included them at the end of the text in the manuscript (after the reference list).
15. We have used the instructions to ensure that the figures are cropped. We did however not see any visible changes to the figures.